Author's response to reviews

Title: Design and Organization of the Dexamethasone, Light Anesthesia and Tight Glucose Control (DeLiT) Trial: A Factorial Trial Evaluating the Effects of Corticosteroids, Glucose Control, and Depth-of-Anesthesia on Perioperative Inflammation and Morbidity From Major Non-cardiac Surgery

Authors:

- Basem Abdelmalak (abdelmb@ccf.org)
- Ankit Maheshwari (mahesha@ccf.org)
- Edward Mascha (maschae@ccf.org)
- Sunita Srivastava (srivass@ccf.org)
- Theodore Marks (markst1@ccf.org)
- WH Wilson Tang (tangw@ccf.org)
- Andrea Kurz (kurza@ccf.org)
- Daniel I Sessler (sessled@ccf.org)

Version: 6 Date: 25 May 2010

Author's response to reviews: see over
Design and Organization of the Dexamethasone, Light Anesthesia and Tight Glucose Control (DeLiT) Trial: A Factorial Trial Evaluating the Effects of Corticosteroids, Glucose Control, and Depth-of-Anesthesia on Perioperative Inflammation and Morbidity From Major Non-cardiac Surgery

Basem Abdelmalak, Ankit Maheshwari, Edward Mascha, Sunita Srivastava, Theodore Marks, WH Wilson Tang, Andrea Kurz and Daniel I Sessler

Dear BMC Editorial Production Team,

Per your request, we have made the following changes to our manuscript.

**Formatting changes to be made:**

<table>
<thead>
<tr>
<th>Major revisions (we require the author to make these changes)</th>
</tr>
</thead>
</table>

**Abstract:** please ensure that the Abstract is present in manuscript and matches exactly what is in the submission system

This has been completed

**Structure:** this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors’ information (optional); Acknowledgements (optional); References.

We formatted the manuscript according to the structure listed above.

**Additional files:** We notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.

We have included the tables in the manuscript following the references.
Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Qualifications: Please remove the authors' qualifications (e.g. PhD / MD) or job titles from the manuscript and submissions system

We removed the author’s qualifications.

Underlining: Please note that underlining is not supported in the final full text; you may wish to change it to bold.

We removed the underlining in the manuscript.

Title page: please remove the short title and word count.

This has been removed.

Abstract: please remove the key words and word count from the abstract.

This has been removed.

Authors contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.

We put the contributions in paragraph format.

References: the reference list should contain up to 30 authors names for each citation. The term et al. should not be used if there are fewer than 30 authors.

We replaced the term et al with the authors names.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have reviewed the manuscript for typographical errors

Additional changes:

Upon through examination of this proof, we identified that the following paragraph was missed and felt that it was important to add it (see page 13). “Blood for high sensitivity CRP (hsCRP) and cardiac troponin T measurements will be sampled at the time of induction, at 1st and 2nd post-operative days”